Author’s response to reviews

Title: A Novel Targeted Angiogenesis Technique Using VEGF Conjugated Magnetic Nanoparticles And In-Vitro Endothelial Barrier Crossing

Authors:

Mark Arokiaraj (christomark@gmail.com)

Version: 1 Date: 27 Oct 2016

Author’s response to reviews:

The authors names are revised. Ebersmedical persons decided to be mentioned in the acknowledgement. This is their decision, and as a first author i have given opportunity and it was their wish.

The study was performed by the first author with scientific interests only.

The methods section have been elaborated and discussion section was modified with more information in this revised manuscript.